Author's response to reviews

Title: Prenatal Risk Factors for Tourette Syndrome: A Systematic Review

Authors:

Ting-Kuang Chao (tkchao@gmail.com)
Jing Hu (jinhu@ucalgary.ca)
Tamara Pringsheim (tmprings@ucalgary.ca)

Version: 3
Date: 8 January 2014

Author's response to reviews:

Response to Reviewer and Editor, Revision 2

1. There are no recommended changes by reviewer 1.
2. We have modified Figure 1 as recommended by reviewer 2, and correction typographical errors.
3. We have included the information on all authors as requested in the title page.
4. We have included the tables in the main manuscript following the references.